Author’s response to reviews

Title: Economic Analyses to Inform Public Health Decision Making for Tuberculosis: The Role of Understanding Implementation

Authors:

Priya Shete (priya.shete@ucsf.edu)

James Kahn (jgkahn@ucsf.edu)

Version: 1 Date: 18 Oct 2019

Author’s response to reviews:

Dear Editors,

Thank you for the opportunity to revise this Commentary. We have made the requested changes and now refer to the original paper by Sohn and colleagues in 2nd paragraph of the Introduction.

Thanks,

Priya Shete MD MPH

(on behalf of the authors)